Author’s response to reviews

Title: Prevalence and Associated Risk Factors of Chronic Bronchitis in First Nations People

Authors:

Punam Pahwa (pup165@mail.usask.ca)
Chandima Karunanayake (chandima.karunanayake@usask.ca)
Donna Rennie (donna.rennie@usask.ca)
Joshua Lawson (josh.lawson@usask.ca)
Vivian Ramsden (viv.ramsden@usask.ca)
Kathleen McMullin (kathleen.mcmullin@usask.ca)
P. Jenny Gardipy (director@willowcreehealth.com)
Judy MacDonald (jjudithlynmacdonald@hotmail.com)
Sylvia Abonyi (sylvia.abonyi@usask.ca)
Jo-Ann Episkenew (firstnations.lung@usask.ca)
James Dosman (james.dosman@usask.ca)

Version: 2 Date: 11 May 2017

Author’s response to reviews:

This has the reference to the email dated April 26, 2017 regarding the manuscript ‘PULM-D-16-00386R1’. Changes have been made as per the requests. We would appreciate if you please make arrangements for the publication of the manuscript at your earliest suitable. Thank you.